Author's response to reviews

Title: In vitro and in vivo anticancer properties of a Calcarea carbonica derivative complex (M8) treatment in a murine melanoma model

Authors:

Fernando de S. F. Guimarães (labbiocel@ufpr.br)
Lucas F. Andrade (lucas_bio@hotmail.com)
Sharon T. Martins (sth_martins@yahoo.com.br)
Reginaldo V. Sene (rvsene@yahoo.com.br)
Ana P. R. Abud (ana_abud@onda.com.br)
Reginaldo V. Sene (rvsene@hotmail.com)
Carla Vanderer (carla.w@ufpr.br)
Inés Tiscornia (inesti@pasteur.edu.uy)
Mariela Bollati-Fogollín (mbollati@pasteur.edu.uy)
Dorly de F. Buchi (buchi@ufpr.br)
Edvaldo S. Trindade (esttrindade@ufpr.br)

Version: 4 Date: 21 March 2010

Author's response to reviews: see over
Editor
BMC Cancer,

Enclosed you will find the actualization of the manuscript entitled “In vitro and in vivo anticancer properties of a Calcarea carbonica derivative complex (M8) treatment in a murine melanoma model” by Fernando de S. F. Guimarães, Lucas F. Andrade, Sharon T. Martins, Reginaldo V. Sene, Carla Wanderer, Inês Tiscornia, Mariela Bollati-Fogolín, Dorly de F. Buchi and Edvaldo S. Trindade as myself being the corresponding author. All authors are aware and support the submission of this manuscript as authors. The follow modifications or observations were done as Reviewer’s suggestions and observations.

Hoping to hear from your soon,
Yours sincerely,

Edvaldo da Silva Trindade, PhD
Professor of Cellular Biology
Laboratório de Pesquisa em Células Neoplásicas e Inflamatórias
Authors names and order were updated in the submission system according the manuscript.